Author’s response to reviews

Title: Awareness of ADHD in primary care: stakeholder perspectives

Authors:

Blandine French (blandine.french@nottingham.ac.uk)
Elvira Perez-Vallejos (mszep@exmail.nottingham.ac.uk)
Kapil Sayal (kapil.sayal@nottingham.ac.uk)
David Daley (david.daley@nottingham.ac.uk)

Version: 2 Date: 13 Feb 2020

Author’s response to reviews:

Dear Gerard Clunn,

Submission of revised paper: Awareness of ADHD in primary care: stakeholder perspectives (FAMP-D-19-00294)

Thank you for potentially accepting our manuscript. We have reviewed the final comments and have revised the manuscript accordingly. Our responses are given in a point by point manner below.

We hope the revised version will satisfy the reviewers and we look forward to hearing back from you in due course.

Sincerely

Blandine French

-- We note that you have included a ‘Consent for publication’ section in the Declarations. Consent for publication refers to consent for the publication of identifying images or other personal or clinical details of participants that compromise anonymity. Seeing as this is not applicable to your manuscript please state “Not Applicable” in this section.
This section has been edited

-- In the section 'Funding', please also describe the role of the funding body/bodies in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

This section has been edited

--Please separate the Discussion and Conclusion sections.

These sections have been separated

-- At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

The manuscript has been edited according to the guide lines above